Author's response to reviews

Title: Associations between e-cigarette access and smoking and drinking behaviours in teenagers

Authors:

Karen Hughes (k.e.hughes@ljmu.ac.uk)
Mark A Bellis (m.a.bellis@ljmu.ac.uk)
Katherine A Hardcastle (k.a.hardcastle@ljmu.ac.uk)
Phil McHale (p.mchale@ljmu.ac.uk)
Andrew Bennett (andrew.bennett@hegroup.org.uk)
Robin Ireland (robin.ireland@hegroup.org.uk)
Kate Pike (k.pike@warrington.gov.uk)

Version: 2 Date: 6 May 2014

Author's response to reviews:

Dear Editor,

Please find attached a revised version of our article which includes line numbers and the full postal address of the submitting/corresponding author.

Many thanks and best wishes
Karen Hughes